

Author's response to reviews

Title: Lewy Bodies and neuronal loss in subcortical areas and disability in non-demented older people: A population based neuropathological cohort study

Authors:

Michelle Byford (m.byford@ucl.ac.uk)
Carol Brayne (carol.brayne@phpc.cam.ac.uk)
Ian McKeith (i.g.mckeith@newcastle.ac.uk)
Mark Chatfield (Mark.Chatfield@mrc-hnr.cam.ac.uk)
Paul G Ince (P.G.Ince@sheffield.ac.uk)
Fiona E Matthews (Fiona.matthews@mrc-bsu.cam.ac.uk)
MRC CFAS (Fiona.matthews@mrc-bsu.cam.ac.uk)

Version: 3 **Date:** 11 May 2009

Author's response to reviews: see over

Institute of Public Health: Cambridge (01223) 330300
Direct line: Cambridge (01223) 330391
Fax: Cambridge (01223) 330388
e-mail: Fiona.matthews@mrc-bsu.cam.ac.uk
http://www.mrc-bsu.cam.ac.uk

Monday, May 11, 2009

Dear Sir / Madam

Thank you for the comments, my apologies about the editing errors. These have now all been corrected.

However, before acceptance, we would urge you to go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at http://www.biomedcentral.com/info/ifora/medicine_journals/

Done as requested

In addition, we have the following editorial requests to make of you:

1. Please remove the following text from your Title Page:

"Disclosure

The authors report no conflicts of interest

Word Count:

Paper: 3830 words

Shortened title: Lewy Bodies and neuronal loss in subcortical areas and disability "

Done as requested

2. Please rename your Introduction section as Background.

Done as requested

3. Please move your ethical statement so that it appears in your Methods section.

Done as requested

4. You must state the Conclusions of your investigation in a dedicated Conclusions section, which should follow your Discussion section:
<http://www.biomedcentral.com/bmcgeriatr/ifora/#conclusions>

Done as requested

5. Please move your Competing Interests and Authors' Contributions sections so that they precede your Acknowledgements section.

Done as requested

Yours sincerely,

Fiona Matthews (Dr)
Senior Research Associate